Author's response to reviews

Title: Mycophenolate and FK506 Have Different Effects on Kidney Allograft Fibrosis in Rats that Underwent Chronic Allograft Nephropathy

Authors:
Lei Luo (luolei7710@yahoo.com.cn)
Lin Zhao Sun (szl7710@hotmail.com)
Heng Guang Luo (luongguangheng1975@hotmail.com)

Version: 2 Date: 18 January 2012

Author's response to reviews: see over
Dear Editor:

We would like to resubmit our accompanying manuscript, entitled “Mycophenolate and FK506 Have Different Effects on Kidney Allograft Fibrosis in Rats that Underwent Chronic Allograft Nephropathy” for publication in the BMC Nephrology.

We have added the name of the ethical committee and a reference number that approved our study in the methods section of manuscript. And also added another author previously omitted.

Finally, we declare that this paper is our original unpublished work, and it has not been submitted to any other journal for review. All authors state that no competing financial interests exist.

We hope that the manuscript now meets the high standard required by the BMC Nephrology. We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Prof. Guangheng Luo
Department of Urology Surgery
Guizhou Province People’s Hospital, Guiyang, 550002, PR China
E-mail address: luoguangheng1975@hotmail.com
Telephone: 86-851-5600886
Fax: 86-851-5600886